Author’s response to reviews

Title: Severe thoracic trauma caused left pneumonectomy complicated by right traumatic wet lung, reversed by extracorporeal membrane oxygenation support—a case report

Authors:

Feng Yun Wang (dr-w@qq.com)
Bin Fang (dr_fangbin@163.com)
Zhi Hui Yu (dr_zhuihu@163.com)
Jing Song Shao (dr_shaojingsong@163.com)
Wei Biao Wen (wwbiao@fsyy.com)
Li Xin Zhou (drzlxin@126.com)

Version: 4 Date: 18 Jan 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your carefulness and guidance!

Dr. Zhou Li-Xin's contribution was written in the authors' contribution section.

The figures were changed back to composite files when necessary.

If we didn't do the revision correctly, please don't hesitate to contact us.

Sorry for our lack of experience!

Thank you for your help!